Author's response to reviews

Title: High prevalence of HIV infection and unprotected anal intercourse among older men who have sex with men in China: A systematic review and meta-analysis

Authors:

Yongze Li (pandawisp@163.com)
Junjie Xu (xjibeijing@gmail.com)
Hanzhu Qian (han-zhu.qian@Vanderbilt.Edu)
Bingxin You (youlong668@hotmail.com)
Jing Zhang (zhangjing1985zi@hotmail.com)
Jianming Zhang (zjm871004@gmail.com)
Qinghai Hu (qinghaihucn@gmail.com)
Zhenxing Chu (aochu323@gmail.com)
Shuyang Liu (liushuyang1987@126.com)
Yongjun Jiang (jiang_yongjun_jc@hotmail.com)
Wenqing Geng (windygeng@163.com)
Hong Shang (hongshang100@hotmail.com)

Version: 3
Date: 1 September 2014

Author's response to reviews:

**********

"Please provide the source to the map image in Figure 4. Could you clarify if you own the copyright for the map image and have permission to reproduce it (and that the copyright holder is aware that the figure will be published in an open access journal under a creative commons licence)."

**********

Response: The map image in Figure 4 was created by Yongze Li, which is one of our co-author. We clarify we own the copyright for the map image and have permission to reproduce it (and that he is aware that the figure will be published in an open access journal under a creative commons licence). Also, we cite the copyright holder and the software used to create the map in the figure legend.